Public Health Effects of Ecosystem Degradation: Evidence from Deforestation in Indonesia *

Teevrat $Garg^{\dagger}$

September 12, 2014

Job Market Paper

Abstract

Despite growing concern about the effect of environmental degradation on human health, little effort has been made to quantify the effect of ecosystem damage on the incidence and burden of infectious diseases. Using village-level panel data and satellite data on forest cover, I find that deforestation between 2001-2008 in Indonesia can explain 360,000-880,000 additional malaria infections. The evidence is consistent with an ecological response and the effect of deforestation on malaria cannot be explained by post-deforestation land use change, anti-malarial programs or migration. The effect is specific to malaria, with deforestation having no discernible effect on other diseases whose disease ecology differs from that of malaria. Back of the envelope calculations suggest that the local health benefits from avoided deforestation are 6-31 times the global carbon benefits underscoring a large, yet previously ignored and unquantified cost of deforestation.

JEL Codes: Q53, Q56, Q57, Q20

^{*}I am extremely grateful for extensive comments and advice from Chris Barrett, Nancy Chau, Shanjun Li and Russell Toth. I thank Ben Olken, Matt Hansen, Indra Surbakti and Samuel Bazzi for sharing Indonesian data. The paper has also benefitted from comments on earlier drafts by Ed Barbier, Antonio Bento, Ariel Ben-Yishay, Jim Berry, Prashant Bharadwaj, Shyamal Chowdhury, Janet Currie, Brian Dillon, Rebecca Edwards, Hayley Fisher, Chris Golden, Chris Goodwin, Jake Hochard, Vivian Hoffmann, Susana Iranzo, Asadul Islam, Ravi Kanbur, Dan Kaffine, Don Kenkel, Joel Landry, Ben Leard, Adrienne Lucas, Matt Neidell, Ben Olken, Evan Plous, Tom Pepinksy, Sharon Poctzer, Alex Reese-Jones, Marc Rockmore, Kevin Roth, Paulo Santos, Kevin Schnepel, David Ubilava, Michael Ward, Will White, Nicolas Ziebarth, seminar participants at Cornell University (Dyson, Policy Analysis & Management), Monash University (Economics), University of Sydney (Economics), 2014 Agricultural and Applied Economics Association's Annual Meetings, 5th World Congress of Environmental and Resource Economists, and the 10th Australasian Development Economics Workshop. Funding is acknowledged from the Cornell Population Center, the Atkinson Center for a Sustainable Future, the Mario Einaudi Center for International Studies, and the Stephen B. and Janice G. Ashley professorship. All opinions and errors are my own.

[†]Email: tg236@cornell.edu. Charles H. Dyson School of Applied Economics & Management, Cornell University

1 Introduction

In addition to the large costs of morbidity and mortality, infectious diseases result in lower economic growth (Acemoglu and Johnson, 2007; Bleakley, 2007; Well, 2007; Gallup and Sachs, 2001), lower educational outcomes (Barreca, 2010; Cutler et al., 2010; Lucas, 2010; Miguel and Kremer, 2004) and declining fertility (Lucas, 2013). The most widespread of these diseases, malaria, affects 300-500 million people each year, half of whose victims are children under the age of five (Lucas, 2010; Sachs and Malaney, 2002), disproportionately affecting poor and vulnerable populations (Ngonghala et al., 2014). Many papers have discussed interventions that can reduce the incidence of such diseases in poor countries (Ashraf, Berry and Shapiro, 2010; Cohen and Dupas, 2010; Hoffmann, 2009; Hoffmann, Barrett and Just, 2009). Yet, even as quantifying the links between environmental factors (eg. air pollution) and human health and well-being has become a central component of the modern research agenda in environmental economics (Graff Zivin and Neidell, 2013; Greenstone and Jack, 2013), little effort has been made to understand the effect of ecosystem degradation on the spread of infectious diseases. This void in research is important to fill since increasingly, interventions dependent on insecticides and anti-malarial drugs are likely to be limited in their effect as a result of increasing resistance of disease-carrying vectors and pathogens, and environmental and natural resource management will be essential in controlling the incidence of infectious diseases (Lindsay and Birley, 2004).

In this paper, I provide the first causal evidence linking deforestation to malaria. In theory, deforestation can affect malaria through several channels. First, post-deforestation land use change in the form of agricultural activity and urbanization has been associated with malaria. Second, migrants can act as latent hosts of malaria since they typically have lower incomes and less access to medical facilities than native populations and deforestation often drives migration in poor countries. Third, deforestation can alter the disease ecology of malaria. In particular, deforestation results in accumulation of water puddles and reduction in biodiversity, both of which favor the growth of anophelene larvae (Pattanayak and Pfaff, 2009; Patz and Olson, 2006; Patz et al., 2000; Walsh, Molyneux and Birley, 1993). While the first two mechanisms are socio-economic in nature, the third is purely ecological. In this paper, I provide evidence that there is a pure ecological response in the effect of deforestation on the incidence of malaria.

Using variation in deforestation and a rich village data set from Indonesia, I find that deforestation has a statistically significant impact on the incidence of malaria ¹. In particular, there are three key findings. First, for the average within-sample loss in forest cover (1000 Ha) in a district, there is a 2% - 4.6% increase in the probability of a malaria outbreak in each village in that district, which translates to about 45,000 - 110,000 additional infected individuals in Indonesia each year. Second, the effect of deforestation on malaria is largely concentrated in villages within or near forests, emphasizing the local nature of the health externalities, in sharp contrast to the global nature of the carbon externalities. Back of the envelope calculations suggest that the health cost of deforestation is between \$123 - \$610 per hectare compared to the carbon benefits of \$20 per hectare². Third, the observed effect is

¹Indonesia is an ideal setting for such a study. It contains the third largest stand of tropical forest in the world experiences the highest rate of deforestation in the world (Margono et al., 2014). Between 2001-2008, Indonesia saw 4.8 million hectares of forest cover loss. In fact, deforestation in Indonesia makes it the third largest producer of greenhouse gasses worldwide, behind the US and China (Burgess et al., 2012). At the same time, malaria poses a major public health risk. The WHO estimates that 44% of the population in Indonesia, over 130 million people, is at risk of malaria infection (Elyazar, Hay and Baird, 2011)

²See Nature (2009); Honorine (2010). The World Bank estimates that Indonesia's 1 million square kilometers of forest cover could earn Indonesia between \$400 million - \$2 billion through carbon credits.

much more pronounced in forests intended to preserve biodiversity, than in forests which are for the purpose of logging and land use change. Furthermore, the effect can not be explained by common reasons provided in the public health literature, such as post-deforestation land use change, anti-malarial programs or migration, suggesting that the underlying mechanism is an ecological as opposed to a socio-economic response.

Despite forests forming a critical part of the environment and ecosystems in poor communities, achieving causal identification of the effect of deforestation on infectious diseases has been difficult (Pattanayak et al., 2006)³. There are two major identification concerns. First, there could omitted variables bias associated with road access, occupational choice⁴, poverty levels⁵, geographic features, sanitation facilities, medical facilities, and agricultural activity. Second, deforestation could be endogenous to malarial outbreak through reverse causality or simultaneity if malarial outbreak affects workers' choice to clear forest through reduced availability of healthy labor.

A major contribution of this paper is to overcome these challenges and provide causal identification of the effect of deforestation on the incidence of malaria by building on important prior research in the public health literature (Bauch, Pattanayak and Sills, 2014; Laporta et al., 2013; Myers et al., 2013; De Silva and Marshall, 2012; Pattanayak and Pfaff, 2009; Yasuoka and Levins, 2007; Patz and Olson, 2006; Patz et al., 2000). I employ panel

That translates to \$4-\$20 per hectare.

 $^{^{3}}$ Reviews of the epidemiology and public health literature by Keesing et al. (2010) and Pongsiri et al. (2009) examine case studies on the links between biodiversity loss and local disease ecology. Both reviews illustrate and recognize that aside from lab experiments that produce limited results, field studies in this area have been largely correlational case studies.

⁴Individuals working in the forestry sector could have increased exposure to vectors in the event of deforestation.

⁵Deforestation could either result in exploitation of local population or provide local economic windfalls and could thus influence vulnerability to diseases through income and access to medical facilities.

methods by exploiting satellite data on forest cover and a rich village census data on disease outbreaks to control for time-invariant geographic, climatic and demographic factors that could be correlated with both deforestation and diseases. This allows me to identify the effect of deforestation on malaria off of the timing of deforestation (and not whether or not a village experienced deforestation) that is plausibly exogenous to local disease outcomes. Administratively most deforestation decisions are made at the district level or at the national level in consultation with district heads.⁶. All forested land in Indonesia is owned by the government, and as such, all legal and illegal deforestation in Indonesia requires the implicit or explicit consent of the district head (*bupati*). Since each district comprised of over 250 villages on average, it is unlikely that decisions to grant licenses or to acquiesce to illegal deforestation by the district head were contingent on malarial outbreaks in specific villages. As such, district-level deforestation in the Indonesian context is plausibly exogenous to village-level disease incidence.

I test this assumption in a few ways. First, I provide placebo tests and show that the effect is specific to malaria. If there were unobservables correlated with health and deforestation, then deforestation would also have an impact on other diseases whose disease ecology differs from that of malaria. Second, I use exogenous variation generated from two national policies: a) the introduction of a timber certification program aimed at curtailing illegal exports of raw logs and b) the national moratorium on district splitting that temporarily halted the creation of new districts, altering the political economy of deforestation (Burgess et al., 2012). Third, I provide falsification tests for these national level policies and show the effects are

⁶Indonesia has many levels of governance. The country is divided into 32 provinces (*propinsi*) which are further divided into districts (*kabupaten*). Districts are subdivided into villages (*desa*). There is also a sub-district (*kecamantan*) level of administration that is responsible for implementing some welfare programs.

specific to the year in which they were implemented. Fourth, I show that the results are robust to the inclusion of a variety of controls including health care access, poverty measures, distance from cities, sanitation facilities, population density, government sponsored health interventions, geographic terrain, precipitation and primary sector of employment.

This paper also contributes to the economics literature on the health and mortality effects of environmental degradation. This literature has made great advances in causal identification of the effect of air pollution on infant mortality (Currie and Walker, 2011; Jayachandran, 2009; Currie and Neidell, 2005; Neidell, 2004; Chay and Greenstone, 2003), adult mortality (Ziebarth, Schmitt and Karlsson, 2014), productivity (Graff Zivin and Neidell, 2012; Hanna and Oliva, 2011) and educational outcomes (Currie et al., 2009).⁷ However, the focus has been almost exclusively on air pollution with much less attention paid to understanding the health effects of other forms of environmental degradation such as deforestation. This paper fills that important void; to the best of my knowledge the economics literature has not previously addressed the causal impact of deforestation on health or disease outcomes.

This research has strong policy implications for payments for ecosystem services (PES) (Jayachandran, 2013; Alix-Garcia et al., 2013; Alix-Garcia, Sims and Yanez-Pagans, 2013; Golden et al., 2011; Alix-Garcia et al., 2009; Andam et al., 2008; Jack, Kousky and Sims, 2008; Angelsen and Wunder, 2003; Foster and Rosenzweig, 2003; Angelsen, 1999; Angelsen and Kaimowitz, 1999; Koop and Tole, 1999; Barbier, 1997; Ferraro and Kramer, 1997).

⁷Several authors have made use of quasi-experimental methods. For instance, Chay and Greenstone (2003) used the geographic variation in air particulate matter induced by the 1981-82 recession to show that reduced air pollution led to decreases in infant mortality to the tune of 2500 fewer deaths per year. Currie and Walker (2011) study the introduction of electronic toll collection to show that reduced pollution through decreased traffic congestion resulted in lower incidence of prematurity and low birth weight amongst mothers within two kilometers of a toll plaza in NJ and PA. Jayachandran (2009) exploits exogenous variation in air pollution from forest fires in Indonesia to demonstrate a large effect of smoke on infant mortality.

First, health benefits exceed carbon benefits, particularly in light of recent evidence that the carbon footprint of deforestation may be as low as 25% of previous estimates (Harris et al., 2012). Second, the health benefits are local in nature, in sharp contrast to the carbon benefits that are global. Since the local health benefits exceed the global carbon benefits, PES programs, such as Indonesia's recently launched Rainforest Standard (RFS), may not require large external funding to avoid deforestation and instead motivate local institutions to promote conservation (JakartaPost, 2014).

This paper is organized as follows. Section 2 provides context for deforestation and the implications for the disease ecology of malaria, and section 3 describes the data. Section 4 discusses the identification strategy. Estimation results are presented in section 5. Section 6 considers alternative explanations and mechanisms and section 7 concludes and draws out policy-relevant findings.

2 Background

While its tropical conditions are ideal for forest growth, Indonesia has been experiencing alarmingly high rates of deforestation. Burgess et al. (2012) provide a conservative estimate that between 2001 and 2008, Indonesia lost about 4.8 million hectares of forest cover, which is roughly the size of the US states of Vermont and New Hampshire combined. This coincides with a period of high rates of contracting infectious diseases. In this section, I provide background on the nature of deforestation as well as its implications for the spread of malaria.

2.1 The Political Economy of Deforestation in Indonesia

During the Suharto era, the national government maintained total control of the forest estate and concessions were awarded to a small group of forestry conglomerates in the production and conversion zones for the purposes of logging and clearing of forested land for industrial and palm oil operations respectively. Since the conglomerates had connections to the Suharto government and because the permits lasted up to 30 years, they had strong incentives to conduct logging selectively and sustainably. Logging was prohibited in all other zones that were designated for watershed protection (protection forest) and biodiversity protection (conservation forest) (Barber, 1990). However, after the fall of Suharto regime, the national government passed the decentralization laws that amongst other things, made district forest departments part of the district government, answerable to the head of the district (Burgess et al., 2012).⁸

60%-80% of all deforestation in Indonesia is illegal (CIFOR, 2004). To a large extent, the onus of authorizing, monitoring and enforcing⁹ limits on logging (legal and illegal) falls on the district office. District office employees are supposed to stationed at the gate of every concession to monitor all logs leaving the concession, and at the entrance of all saw mills to check the legality of logs entering the mill. Furthermore, all logs require transport permits from the district office that are verified at regular checkpoints. Thus the extraction, transportation or processing of illegal logs would be impossible without the district office

 $^{^{8}}$ The district governments also became incredibly powerful. They were granted considerable autonomy over their jurisdiction and 25% of revenues from natural resources in a district were returned to the district governments as block grants.

⁹Burgess et al. (2012) demonstrate that illegal deforestation can not be explained by lack of enforcement capacity since the impact of new districts on deforestation rates increases over time. If lack of enforcement capacity was driving deforestation, the effect of new districts would decrease over time since newer districts would be able to build up enforcement capacity.

acquiescing to such activity (Burgess et al., 2012). In a production forest, the district head may issue permits for logging up to the point of the nationally sanctioned limit, and has the responsibility to ensure that only legal deforestation occurs. In the case of conversion forests, the district government petitions the central government to convert forest land to other uses, such as palm oil plantations, and is responsible for monitoring conversion to ensure that it occurs only in approved areas. In protection and conservation forests, all deforestation activity is illegal. These forests are biodiversity reserves and come under the control of the central government. But the district office is responsible for patrolling these forests to ensure no illegal logging takes place. The ecology of production/conversion forests is distinct from that of protection/conservation forests in part due to historical logging, but the political economy of deforestation is largely similar across the different forest zones.

2.2 Implications for Malaria

Malaria is spread only through the *Anopheles* mosquito and transmission occurs only if the mosquito was previously infected through a blood meal taken from an infected person¹⁰. Subsequently the malarial parasites from the first blood meal are mixed in with the saliva of the mosquito and then transmitted into the person being bitten. The typical lifespan of the female *Anopheles* is 4-6 weeks, but they can travel distances as far as 75 miles, making the the range of transmission particularly problematic. In Indonesia, malaria is widespread with 44% of the population, or over 130 million people, at risk of contraction (Elyazar, Hay and Baird, 2011). In any given year, 19% of Indonesian villages report having an outbreak of malaria.

¹⁰See: http://www.cdc.gov/malaria/about/faqs.html

Pattanayak and Pfaff (2009) review potential mechanisms through which deforestation can impact the growth of anopheline larvae and consequently the incidence of malaria. First, post-deforestation land use change in the form of urbanization, construction and agricultural production has been associated with increased incidence of malaria (De Silva and Marshall, 2012; Petney, 2001). Second, deforestation commonly results in migration, which often increases the incidence of malaria for two reasons - migrants act as latent hosts for infectious diseases, acting as transport from an infected area to a non-infected area (Texier et al., 2013) and migrants tend to have limited access to medical facilities. These are socio-economic factors behind the link between deforestation and malaria but there is also an ecological link. Deforestation also alters the disease ecology of malaria. This occurs in two ways. First, cleared lands receive more sunlight and are more susceptible to the formation of puddles with a more neutral pH that favors anopheline larvae development (Pattanayak and Pfaff, 2009; Patz and Olson, 2006; Patz et al., 2000; Walsh, Molyneux and Birley, 1993). Second, deforestation adversely affects biodiversity of the region and increases malaria incidence by favoring the proliferation of malaria related species by reducing or eliminating species that prev on anopheline larvae. The evidence is bi-directional; increased biodiversity also reduces malaria incidence through dilution of disease carrying vectors in the overall species pool in forest ecosystems (Laporta et al., 2013; Yasuoka and Levins, 2007).

3 Data

3.1 Disease and Public Health

The Indonesian statistical agency, *Badan Pusat Statistik* (BPS) conducts a village census *Podes* every 2-3 years that documents a range of village characteristics and events including the incidence and extent of disease outbreaks for malaria, measles, respiratory infections, dengue fever and diarrheal diseases. The survey includes all of Indonesia's over 68,000 villages. During the period of study, *Podes* was conducted in 2000, 2003, 2006, and 2008. Village heads report on whether or not there had been an outbreak of each of these diseases in their village, and if there was an outbreak how many people died. For 2000, 2003 and 2006, no information was collected on the number of people who were infected with malaria. Starting in 2008, the survey also requested information on the number of people infected with malaria. In 2008, on average, 14-15 people were infected with malaria in a village that reported an outbreak of malaria. ¹¹ The standard deviation was 20 people. This translated to just over 15 infected persons per thousand persons with a standard deviation of 26 infected persons per 1000 persons.

Podes also contains information on the availability of medical facilities including hospitals, integrated health centers, village medicine posts and drug stores in a given village. When such a medical facility is not present in the village, the survey documents the distance to the nearest facility (in kilometers) and how easy it is to access such a facility on a scale of 1-4. This rich data set also includes information on the source of drinking water, sanitation

¹¹The data is collected through oral recall from village heads and is then verified at the sub-district office. The village head makes a subjective decision on whether or not there was an outbreak in a given year. A kernel density graph of the number of infected individuals conditional on an outbreak is available in the appendix.

facilities, transportation facilities, land utilization, geography (proximity to the shore, forests etc.) and topography (flat, hilly or valley). Tables 1 - 4 show the summary statistics of the key dependent and independent variables.

3.2 Deforestation Data

Since a large portion of the deforestation in Indonesia results from illegal logging, official records are likely inaccurate. Therefore, I use deforestation data from 2001-2008 based on satellite imagery data directly from Burgess et al. (2012). Deforestation is measured as the number of pixels (250m X 250m) in each district estimated from MODIS satellite images that have at least a 90% probability of tree cover loss compared to the previous year. The variable is converted into 1000s of hectares. While the Indonesian archipelago is comprised of a large number of islands, we focus on the four islands that have substantial forest cover: Sumatra, Kalimantan, Sulawesi and Papua. The remaining islands are excluded since they had negligible forest cover in the baseline year of the study (2000). The key data improvement from previous economic research on deforestation using satellite data (Foster and Rosenzweig, 2003) is building on the Normalized Difference Vegetation Index (NDVI) (Tucker, 1979) by also making use of differences in spectral signatures between trees and other crops (Jensen, 1996). The deforestation data are then classified by the zone and district of the forest estate into production, conversion, protection or conservation forest (categories defined below). The resulting variable is $TotalDeforestation_{zdt}$ which measures thousands of hectares deforested in forest zone z, in district d at time t.

4 Estimation and Identification Strategy

There are two key parameters of interest. First, what is the effect of deforestation on the probability of malarial outbreak (i.e effects at the extensive margin)? Second, what is the effect of deforestation on the number of malaria deaths conditional on the village having an outbreak of malaria (i.e effects at the intensive margin)?¹² This section explains my strategy for generating plausible estimates of these causal effects.

4.1 Extensive Margin

Since the primary dependent variable is binary (whether or not there was a disease outbreak in the past year), I use a linear probability model¹³ in the baseline specification. The primary specification is,

$$Pr[disease_{vdt} = 1] = \beta_0 + \beta_1 TotalDeforestation_{dt} + \beta_2 X_{vdt} + \mu_d + \eta_{it} + \epsilon_{vdt}$$
(1)

The dependent variable $disease_{vdt}$ denotes whether or not there was a disease outbreak in village v in district d at time t and $TotalDeforestation_{dt}$ is the total deforestation in district d at time t, a continuous variable measured in thousands of hectares. X_{vdt} is a vector of controls including access to medical facilities, terrain and geography, altitude, sanitation

¹²Ideally, the intensive margin of study would be the number of individuals infected. However, those data are unavailable for all years prior to 2008, and so the village-level intensive margin being studied is the number of individuals dying from malaria. These characterizations would be different at the individual level, where a person getting infected would mark the extensive margin, and how much they suffer as result of malaria, including mortality, would mark the intensive margin.

 $^{^{13}}$ In the appendix, I demonstrate the robustness of the baseline results to the use of logit or probit models in place of a linear probability model. I further demonstrate that only 3% of all predicted values lie outside the feasible [0,1] bounds under the LPM. My results are robust to the exclusion of these data points.

and trash disposal facilities, dominant fuel source, rainfall average and rainfall variation, dominant occupation in the village as well as the interaction effects of occupation-year, occupation-province fixed effects, forest zone, forest zone-year and forest zone-province fixed effects. These controls are not included in the primary specification (1) but are included in extensions to the primary specification (Table 5, Columns 3-4). μ_d is the district-level fixed effect, η_{it} is the island - year fixed effect and ϵ_{vdt} is the regression residual. Standard errors are clustered at the district level to allow for arbitrary correlation within a district and over time accounting for spatial autocorrelation and serial correlation. I also include population density (population/total area) and number of other villages in the district that had an outbreak of malaria in that year to control for size of district effects and possible spill-over effects.

 β_1 (extensive margin) is the parameter of interest, and there are two problems in its consistent identification: a) omission of key within-district time-varying variables that are correlated with both deforestation and malaria may generate omitted variable bias, and b) malaria may be a factor in the decision to cut down trees resulting in reverse causality or simultaneity bias. The challenge is finding a plausibly exogenous source of variation in deforestation that minimizes these prospective sources of bias, without removing crucial context that is necessary to test for a biophysical mechanism and provide policy-relevant results. Identification is achieved using panel data methods using district fixed effects and island-year fixed effects, relying on the plausible assumption that the timing of deforestation is exogenous to local disease outcomes conditional on observables.

This assumption is justified for a few reasons. First, and as discussed in the previous section, all legal and illegal deforestation in Indonesia requires the implicit or explicit consent

of the district government. Each district comprises over 250 villages, on average, making it unlikely that decisions by the district head to grant licenses or acquiesce to illegal deforestation were contingent on malarial outbreaks in specific villages. Second, the use of satellite deforestation data at the 250mX250m resolution means that any smallholder deforestation would have to be both large enough to be noticed and agreed upon by village heads and district governments. That too seems very unlikely. Third, the process of obtaining permits or getting authorization on plans to convert forest land from the national government is a cumbersome bureaucratic process which can last several years between when a permit is requested and when it is granted. Fourth, there were two major national level policies that altered the political economy of deforestation: a) the introduction of a timber certification program aimed at curtailing illegal exports of raw logs and b) the national moratorium on district splitting that temporarily halted the creation of new districts (Burgess et al., 2012). It is highly unlikely that these policies were introduced as a result of local level disease outbreaks.

Furthermore, I test these assumptions in a few ways, and the results of these tests are discussed in section 5. First, I use placebo tests by estimating the effect of deforestation on other diseases whose disease ecology differs from that of malaria. If there were unobservables correlated with health and deforestation, then deforestation would also have an impact on other diseases whose disease ecology differs from that of malaria. Second, I show that the interaction of deforestation with the timing of these national policies is driving the core result. Third, I provide falsification tests for these national level policies and show the effects are specific to the year in which they were implemented. Fourth, I show that the results are robust to the inclusion of a range of control variables that are known to impact malaria including health care access, poverty measures, distance from cities, sanitation facilities, population density, government sponsored health interventions, geographic terrain, precipitation and primary sector of employment (Pattanayak and Pfaff, 2009).

4.1.1 Logging Ban Enforcement

In the post-Soeharto era, between 1998 and 2001, Indonesia reduced export taxes on logs from 200 percent to 10 percent, resulting in massive exports of logs. At the end of 2001, Indonesia banned all log exports and in late 2003 introduced the SVLK (Timber Legality Verification System) certification to ensure the legality of traded timber (JakartaPost, 2013). Since the policy and enforcement was nationwide, I interact the enforcement of the log export ban with deforestation to introduce exogenous variation in deforestation driven by a national policy. The resulting specification is,

$$Pr[disease_{vdt} = 1] = \beta_0 + \beta_1 TotalDeforestation_{dt} * PostLogExportBan_t + \beta_2 TotalDeforestation_{dt} + \beta_3 X_{vdt} + \mu_d + \eta_{it} + \epsilon_{vdt}$$
(2)

Since the policy was implemented nationwide, identification of β_1 in this specification rests on the assumption that the timing of the implementation of the national policy was uncorrelated with differential levels of malarial prevalence among districts within an island in Indonesia. Specifically, when we include both the interaction term, and deforestation in the regression, finding a statistically significant β_1 estimate and a statistically insignificant β_2 estimate would suggest that identification is being driven by exogenous variation in deforestation following the enforcement of a national policy impacting the forest sector.

4.1.2 Moratorium on District Splitting

In 1999, Indonesia passed the decentralization laws that came into effect in the beginning of 2001. The dramatic decentralization, in the wake of the end of the Soeharto era, provided substantial autonomy to district governments, including returning 25% of all natural resource revenues as block grants. In addition, districts in Indonesia were split or carved into smaller districts, with each district having its own elected district head. In 2008 (2000) Indonesia had 483 (292) districts, with 312 (189) districts in 2008 (2000) in the four major islands included in this study. However, between 2004 and 2007 there was a national moratorium on further district splitting. Burgess et al. (2012) show that this district splitting increased the competition amongst district heads to provide logging rights (legal and illegal). Therefore, I create indicator variables for the start and the end dates (2004 and 2007 respectively) of the moratorium and individually interact those indicator variables with deforestation and add the resulting variables to the original specification in equation (1).¹⁴ The resulting specification is,

$$Pr[disease_{vdt} = 1] = \beta_0 + \beta_1 TotalDeforestation_{dt} * MoratoriumStart_t + \beta_2 TotalDeforestation_{dt} * MoratoriumEnd_t + \beta_3 TotalDeforestation_{dt} + \beta_4 X_{vdt} + \mu_d + \eta_{it} + \epsilon_{vdt}$$
(3)

Since the moratorium was implemented nationwide, identification of β_1 and β_2 in this specification rests on the assumption that the timing of the implementation of the national policy was uncorrelated with differential levels of malarial prevalence among districts within an

 $^{^{14}}$ MoratoriumStart = 1 between 2004 and 2007, MoratoriumEnd = 1 after 2007

island in Indonesia. Specifically, when we include the interaction terms, and deforestation in the regression, finding a statistically significant β_1 or β_2 estimate and a statistically insignificant β_3 estimate would suggest that identification is being driven by exogenous variation in deforestation following the moratorium.

4.2 Intensive Margin

The second variable of interest is the number of malaria deaths in a given year conditional on there being an outbreak of malaria in that village. I estimate an equation similar to (1),

$$MalariaDeaths_{vdt} = \gamma_0 + \gamma_1 TotalDeforestation_{dt} + \gamma_2 X_{vdt} + \mu_d + \eta_{it} + \epsilon_{vdt}$$

$$(4)$$

where γ_1 is the parameter of interest, *MalariaDeaths* is the number of malaria deaths in village v of district d at time t, and as in the case of (1) I estimate the fixed effects model.¹⁵ In addition, since the number of malaria deaths is a count variable, I check the robustness of the fixed effects estimates of (4) to re-estimation using a fixed effects poisson model estimated via maximum likelihood and to a fixed effects negative binomial model.

5 Results

The main result of this paper is that deforestation increases malarial incidence. The central result is that the average within sample deforestation of 1000 hectares in a district increases

 $^{^{15}}$ I run the same specification with "malaria deaths per 1000 persons" as the dependent variable. The results are qualitatively similar, and reported in the appendix.

the probability of malarial outbreak in each village in that district by 0.36 percentage points (Table 5, column 4). I treat this as the baseline result. Using the sample average of the probability of malarial outbreak of 0.19, I find that 1000 hectares of deforestation in a district increases the probability of malarial outbreak in *each* village in that district by 2%. The effect is robust to the inclusion of a wide range of controls and to alternative estimation methods. This estimated effect in the robustness checks (table 7) range from 3.3% - 4.6% suggesting that 1000 hectares of deforestation in a district increase the probability of malaria outbreak in *each* village in that district of malaria outbreak in *each* village in that 1000 hectares of deforestation in a district increase the probability of malaria outbreak in *each* village in that district by 2% - 4.6%.

This range of effects on malaria outbreak, using back of the envelope calculations and average infections per outbreak from the 2008 Podes data, translate to 45,000 - 110,000 additional infections each year. Over the 8-year course of the study that is 360,000 - 880,000 additional person-year infections. I then use estimates on average time loss due to malariainduced morbidity (1.5-3 weeks) and measures on value of statistical life (\$3.7 million) to estimate the morbidity associated health cost per hectare of deforestation in the range of \$123 - \$610. These estimates are 6-31 times the carbon benefit associated with avoided deforestation (\$20 per hectare).

5.1 Placebo Tests

Different diseases have different disease ecologies and as such different mechanisms of transmission. For instance, measles is an airborne disease, while diarrhea is a water-borne disease. I use incidence data on other diseases to perform placebo tests of the core result and the underlying mechanism. In particular, I explore the relationship between deforestation and four other diseases and classes of diseases, namely dengue, measles, diarrhea and respiratory diseases using specification (4) from table 5. As shown in table 6, there is no discernible effect of deforestation in the previous year on the incidence of the other diseases. The lack of any observable effect of deforestation on dengue, measles, diarrhea and respiratory diseases suggests that omitted variables, that could be correlated with both deforestation and health, are not a threat to the internal validity of the central result or the underlying biophysical mechanism. The lack of a discernible effect of deforestation on dengue is also spread via disease carrying vectors, particularly the *Aedes Aegypti* mosquito. However, the *Aedes Aegypti* mosquito, unlike the female *Anopheles* mosquito that carries malaria, has a lifespan of only 2-4 weeks and travels a maximum of $500m(<1/3 \text{ of a mile}; 1 \text{ hectare} = 10,000m^2)$ (*Anopheles* can travel up to 75 miles) during its entire lifetime. This short lifespan and travel-span limit the mechanism of exposure to dengue since the average within-sample deforestation in this study is just over 1000 hectares per district.

The placebo tests underscore the consistent identification of β_1 in equation (1), since any upward bias generated as a result of omission of variables correlated with human health and deforestation (eg. political economy variables such as corruption, favoritism, etc.) would also show up in the effect of deforestation on other diseases.

5.2 Falsification Tests

In table 7, I demonstrate the robustness of the central result presented in table 5. First, the interaction of deforestation with the enforcement of the log export ban has a 0.87 percentage

point effect on the probability of malarial outbreak. Within the sample of this study, the average deforestation in a district (following the logging ban enforcement) increases the probability of malarial outbreak by an estimated 4.56% in every village in that district. The coefficient estimate on deforestation uninteracted is statistically insignificant. The credibility of this result is further strengthened when a "false" year for the enforcement of the log export ban is used instead of the actual year. Specifically, I create a dummy variable that dummy variable with deforestation. The interaction term's coefficient becomes statistically indistinguishable from zero, and the deforestation coefficient has a remarkably similar point estimates to the fixed effects result from table 5 and is statistically significant at the 5% level (table 7, column 2).

Next, I use the start and end dates of the district splitting moratorium (2004 and 2007 respectively) to introduce exogenous variation in deforestation. The coefficient of deforestation interacted with the start of the moratorium is similar to the core fixed effects result and is statistically significant at the 1% level (table 7, column 3). Finally, there are no lead or lagged effects (one lead, one lag) and I find no evidence for non-linearity (results for various non-linear specifications are shown in the appendix).

5.3 Type of Forest and Proximity to Forest

In this sub-section, I provide evidence that the underlying mechanism linking deforestation to malarial outbreak appears ecological and is likely driven by reduced biodiversity. First, I disaggregate results by the type of forest in columns (1) and (2) of table 8. As mentioned in the section 2.1, production and conversion forests are where legal (and illegal) deforestation take place and are demarcated as such for the purposes of logging and land use change, respectively. Conversely, conservation and protection forests are designated as biodiversity reserves where all deforestation is illegal. Upon disaggregating the results by the type of forest, I find that the effect is largely concentrated in protection and conservation forests and is four times larger (0.0136 percentage points v. 0.00149 percentage points) than the effect when all forest types are considered. If the underlying mechanism was socio-economic then there would be no reason to expect different effects across different forest types since the major distinction between the forests is that one set is meant for (managed) clearing and the other set is demarcated for biodiversity protection. Because there is a substantial difference in the magnitude (and statistical significance) of the effect on malaria of deforestation across these different forest types, it is likely that the underlying mechanism is an induced ecological response.

Second, I disaggregate results by villages that are classified as "inside" and "on the border" of forests from villages classified as being "away" from forests ¹⁶. These results are shown in column (3) and (4) of table 8. As is consistent with the maintained hypothesis, the effect is concentrated in villages inside or near forests. The effect in villages away from the forest is statistically indistinguishable from zero and less than half the magnitude of the estimated effect inside or near forests (0.00647 percentage points v. 0.00291 percentage points). This result supports the hypothesized mechanism since deforestation induced *Anopheline* growth is likely to have a larger impact on villages inside or near forests than those further

¹⁶This characterization of the location of the village relative to the forest is provided by village heads and then verified at the district or sub-district office.

away from them because of proximity to growth sites.

5.4 Intensive Margin

The nature of the secondary data used in this paper allows us to study the effects of deforestation on malaria at the both the extensive and intensive margins. Deforestation, if it alters the local disease ecology of malaria, should change the rates of exposure and hence the likelihood of contracting malaria. However, once a person or group of persons have contracted malaria, changes in vector concentration should not impact whether a person survives malaria since that would depend on availability and access to medical facilities. Therefore, if the underlying mechanism linking deforestation to malaria is an ecological one, then deforestation should have a direct impact on the contraction of infection (extensive margin) but conditional on medical facilities and other covariates, should not have a sizeable impact on the number of malaria related deaths (intensive margin).

Consistent with this hypothesis, I find a robust result at the extensive margin (as reported in section 5.2) but no economically meaningful effect at the intensive margin (table 9). The absence of evidence of any effect is consistent across the range of fixed effects models and the poisson (table A.5, column 1) and negative binomial models (table A.5, column 2).

6 Alternative Explanations

In theory, there are three other plausible mechanisms could drive the statistical relationship between deforestation and the incidence of malaria: 1) urbanization and land use postdeforestation drive the incidence of malaria, 2) less remote places with no deforestation are more likely to receive anti-malarial and insecticide treated nets (ITNs) than those with deforestation, and 3) increased migration correlated with deforestation could lead to higher prevalence of malaria since migrants can act as latent hosts of malaria. In this section, I examine these mechanisms as a test of the internal validity of the core result.

6.1 Post-deforestation land use change

A threat to the internal validity of the core result and the underlying mechanism could be that it is not deforestation that causes an increased likelihood of malarial outbreak, but rather the urbanization or land use change (e.g. agriculture, palm oil plantations, etc.) that follows the clearing of forested land (Keesing et al., 2010). This study takes advantage of the rich data in *Podes* and the nature of deforestation in Indonesia to rule out this possibility in three ways. First, I isolate results based on the type of forests being cleared. If postdeforestation land use change drives the relationship between deforestation and malaria, we would observe the effect in production and conversion forests (where forest land can be legally, with permits, cleared for logging and land use change) and not in protection and conservation forests (where forest clearing is illegal but still occurs; see Burgess et al. (2012)). Table 8 shows the exact opposite when I break down the results by forest zoning.

Second, the inclusion of a large number of controls and dummy variables that are likely correlated with land use change does not alter the central result. In particular, I control for area under rice cultivation, total forest cover in a given year and dominant occupation (disaggregated within agriculture) fixed effects as well as occupation-province and occupation-year fixed effects.¹⁷ This helps remove any spurious correlation that was driven by otherwise

¹⁷The dominant occupation of a village was one of eleven different options - crops and plantations, livestock,

unobserved changes in occupational choices in villages, as well as any idiosyncratic changes to these occupations in specific provinces or years.

Finally, if urbanization drives the results, we would expect a direct effect on the outbreak of dengue as well since dengue disproportionately affects urban areas and risk of dengue increases with degree of urbanization (Bhatt et al., 2013). The lack of any effect on dengue prevalence suggests that urbanization can not explain the effect of deforestation on increased risk of malarial incidence.

6.2 Correlation between anti-malarial programs and deforestation

If there is correlation between anti-malarial programs and deforestation in a spatially systematic manner, in that presumably less remote places with no forests are more likely to receive anti-malarial devices, services and programs, then the fixed effects estimator will generate inconsistent results. I address this concern in two ways. First, I disaggregate results by proximity of villages to the forest (table 8, columns 3-4). When considering only the set of villages inside or near forests(table 8, column 4), the results are stronger (both in magnitude and statistical significance) suggesting that such a correlation, if it exists, at worst downward biases the central result. Second, the central result is robust to controlling for percentage (table 5, columns 3-4) of the population in each village that receives some form of welfare payment (poor cards) or subsidized medical care (health cards) making it unlikely that such a spatially systematic correlation exists that would upward bias the main finding of this paper.

inland fisheries, marine fisheries, forestry, other agriculture, mining, industry, wholesale/retail, services and other occupations. Data on percentage of population employed in each sector were unavailable.

6.3 Migration

A concern with the internal validity of the central result and the underlying ecological mechanism is that increased migration correlated with deforestation could lead to higher prevalence of malaria because migrants can act as latent hosts of malaria (Texier et al., 2013). This can be ruled out for a few reasons. First, the placebo tests demonstrate that the effect is specific to malaria; migrants working in deforestation would have to be carrying malaria and no other disease. Second, I explicitly test for migration using population and population density as control variables and find that the result doesn't change. However, migration could still bias the core result if conditional on migration rates, migrants to villages experiencing deforestation are more likely to carry malaria than those to villages not experiencing deforestation. This concern can be ruled out for a few reasons. First, the core result remains robust to the inclusion of the dominant occupation in a village and occupation-year and occupation-province fixed effects. These allow me to control for changes in dominant occupation specific to a particular province or year that could be correlated with migration and deforestation.

Second, if the migration hypothesis is true, districts with production and conversion forests should attract a larger migrant population than districts with protection and conversion forests since average deforestation in production and conversion forests is substantially (five times) higher than in protection and conversion forests. Thus, if migration were the underlying mechanism then production and conversion forests would see larger effects of deforestation on malaria since they attract more deforestation related migrants. Yet, when the results are disaggregated by forest zone, the effect is substantially stronger in protection and conservation forests, making it unlikely that migration is driving the effect of deforestation on increased malaria incidence.

7 Conclusion

Every year over 300 million people are affected by malaria and other infectious diseases resulting in losses in productivity, labor supply, and education as well as increased human suffering. This paper provides causal evidence on the effect of deforestation on the incidence of malaria. I find that using within-sample averages, the mean deforestation in a district in a given year resulted in a 2% - 4.56% increase in the incidence of malaria in every village in that district. I rule out post-deforestation land use change, anti-malarial programs and migration as possible explanations for this effect, highlighting the likely ecological response that drives the effect.

Given the lack of consensus in the political institutions towards action on climate change, this paper provides further motivation for forest conservation and in particular strong evidence of the direct human health impacts of deforestation. First, the absolute numbers are large; deforestation in Indonesia over the 8 year period of this study can explain 360,000 - 880,000 additional infected individuals. This represents a sizeable human health cost of deforestation to the tune of \$123 - \$610 per hectare. This is substantially larger than the carbon benefit of avoided deforestation of \$20 per hectare. Second, the effect seems more pronounced in villages closer to forests that those further away highlighting that there are indeed local benefits to forest conservation in addition to the more often discussed global carbon benefits. This is particularly important in light of Indonesia's adoption of the Rainforest Standard (RFS), a payment for ecosystem services program. Third, the concentration of the effect in protection and conservation forests means that preservation of forest reserves provides health benefits to local communities. However, these are short term results and provide motivation for research in to the long term relationship between natural resource degradation and local public health.

References

- Acemoglu, Daron, and Simon Johnson. 2007. "Disease and Development: The Effect of Life Expectancy on Economic Growth." *Journal of Political Economy*, 115(6): pp. 925–985.
- Alix-Garcia, Jennifer, Alain de Janvry, Elisabeth Sadoulet, and Juan Manuel. 2009. "Lessons learned from Mexico's payment for environmental services program." In Payment for Environmental Services in Agricultural Landscapes. 163–188. Springer.
- Alix-Garcia, Jennifer, Craig McIntosh, Katharine RE Sims, and Jarrod R Welch. 2013. "The ecological footprint of poverty alleviation: evidence from Mexico's Oportunidades program." The Review of Economics and Statistics, 95(2): 417–435.
- Alix-Garcia, Jennifer M, Katharine RE Sims, and Patricia Yanez-Pagans. 2013. "Only One Tree from Each Seed? Environmental effectiveness and poverty alleviation in programs of payments for ecosystem services." Working paper, University of Wisconsin, Madison, Department of Agricultural and Applied Economics.
- Andam, Kwaw S., Paul J. Ferraro, Alexander Pfaff, G. Arturo Sanchez-Azofeifa, and Juan A. Robalino. 2008. "Measuring the effectiveness of protected area networks in reducing deforestation." *Proceedings of the National Academy of Sciences*, 105(42): 16089– 16094.
- **Angelsen, Arild.** 1999. "Agricultural expansion and deforestation: modelling the impact of population, market forces and property rights." *Journal of Development Economics*, 58(1): 185–218.
- Angelsen, Arild, and David Kaimowitz. 1999. "Rethinking the causes of deforestation: lessons from economic models." *The World Bank Research Observer*, 14(1): 73–98.
- Angelsen, Arild, and Sven Wunder. 2003. "Exploring the Forest Poverty Link." CIFOR occasional paper, 40: 1–20.
- Ashraf, Nava, James Berry, and Jesse M Shapiro. 2010. "Can Higher Prices Stimulate Product Use? Evidence from a Field Experiment in Zambia." *American Economic Review*, 100: 2383–2413.
- Barber, C.V. 1990. "The Legal and Regulatory Framework and Forest Production in Indonesia." In Community Land Rights, Customary Law, and the Law of Timber Concessions in Indonesias Forests: Legal Options and Alternatives in Designing the Commons., ed. C. Zerner. Jakarta, Indonesia: Ministry of Forestry/FAO.
- **Barbier, Edward B.** 1997. "Introduction to the environmental Kuznets curve special issue." *Environment and Development Economics*, 2(04): 369–381.
- Barreca, Alan I. 2010. "The long-term economic impact of in utero and postnatal exposure to malaria." *Journal of Human Resources*, 45(4): 865–892.

- Bauch, Simone C., Subhrendu K Pattanayak, and Erin O Sills. 2014. "Public health impacts of protected areas and roads in the Amazon." *Working Paper*.
- Bhatt, Samir, Peter W Gething, Oliver J Brady, Jane P Messina, Andrew W Farlow, Catherine L Moyes, John M Drake, John S Brownstein, Anne G Hoen, Osman Sankoh, Monica F. Myers, Dylan B. George, Thomas Jaenisch, G.R. William Wint, Cameron P. Simmons, Thomas W. Scott, Jeremy J. Farrar, and Simon I. Hay. 2013. "The global distribution and burden of dengue." Nature.
- Bleakley, Hoyt. 2007. "Disease and development: evidence from hookworm eradication in the American South." *Quarterly Journal of Economics*, 122(1): 73–117.
- Burgess, Robin, Matthew Hansen, Benjamin A Olken, Peter Potapov, and Stefanie Sieber. 2012. "The Political Economy of Deforestation in the Tropics." *Quarterly Journal of Economics*, 127(4): 1707–1754.
- Chay, Kenneth Y, and Michael Greenstone. 2003. "The impact of air pollution on infant mortality: evidence from geographic variation in pollution shocks induced by a recession." *Quarterly Journal of Economics*, 118(3): 1121–1167.
- **CIFOR.** 2004. "Generating Economic Growth, Rural Livelihoods, and Environmental Benets from Indonesias Forests: A Summary of Issues and Policy Options." Center for International Forestry Research, report prepared for the World Bank.
- Cohen, Jessica, and Pascaline Dupas. 2010. "Free Distribution or Cost-Sharing? Evidence from a Randomized Malaria Prevention Experiment." Quarterly Journal of Economics, 125(1): 1–45.
- Currie, Janet, and Matthew Neidell. 2005. "Air pollution and infant health: What can we learn from California's recent experience?" *Quarterly Journal of Economics*, 120(3): 1003–1030.
- Currie, Janet, and Reed Walker. 2011. "Traffic Congestion and Infant Health: Evidence from E-ZPass." American Economic Journal: Applied Economics, 3(1): 65–90.
- Currie, Janet, Eric A Hanushek, E Megan Kahn, Matthew Neidell, and Steven G Rivkin. 2009. "Does pollution increase school absences?" The Review of Economics and Statistics, 91(4): 682–694.
- Cutler, David, Winnie Fung, Michael Kremer, Monica Singhal, and Tom Vogl. 2010. "Early-life malaria exposure and adult outcomes: Evidence from malaria eradication in India." *American Economic Journal: Applied Economics*, 72–94.
- De Silva, Prathiba M, and John M Marshall. 2012. "Factors contributing to urban malaria transmission in sub-Saharan Africa: a systematic review." Journal of Tropical Medicine, 2012.
- Elyazar, Iqbal RF, Simon I Hay, and J Kevin Baird. 2011. "Malaria distribution, prevalence, drug resistance and control in Indonesia." Advances in Parasitology, 74: 41.

- Ferraro, Paul J, and Randall A Kramer. 1997. "Compensation and economic incentives: reducing pressure on protected areas." Last stand: protected areas and the defense of tropical biodiversity. Oxford University Press, New York, 187–211.
- Foster, Andrew D, and Mark R Rosenzweig. 2003. "Economic growth and the rise of forests." *Quarterly Journal of Economics*, 118(2): 601–637.
- Gallup, John L., and Jeffrey D. Sachs. 2001. "The Economic Burden of Malaria." American Journal of Tropical Medicine and Hygiene, 64(1-2): 85–96.
- Golden, Christopher D., Lia C. H. Fernald, Justin S. Brashares, B. J. Rodolph Rasolofoniaina, and Claire Kremen. 2011. "Benefits of wildlife consumption to child nutrition in a biodiversity hotspot." *Proceedings of the National Academy of Sciences*, 108(49): 19653–19656.
- Graff Zivin, Joshua, and Mathew Neidell. 2013. "Environment, Health, and Human Capital." *Journal of Economic Literature*, 51: 689–730.
- Graff Zivin, Joshua, and Matthew Neidell. 2012. "The impact of pollution on worker productivity." *American Economic Review*, 102: 3652–3673.
- Greenstone, Michael, and B Kelsey Jack. 2013. "Envirodevonomics: A Research Agenda for a Young Field." *NBER Working Paper*, , (w19426).
- Hanna, Rema, and Paulina Oliva. 2011. "The effect of pollution on labor supply: Evidence from a natural experiment in Mexico City." National Bureau of Economic Research.
- Harris, Nancy L., Sandra Brown, Stephen C. Hagen, Sassan S. Saatchi, Silvia Petrova, William Salas, Matthew C. Hansen, Peter V. Potapov, and Alexander Lotsch. 2012. "Baseline Map of Carbon Emissions from Deforestation in Tropical Regions." Science, 336(6088): 1573–1576.
- Hoffmann, Vivian. 2009. "Intrahousehold allocation of free and purchased mosquito nets." American Economic Review, 99(2): 236–241.
- Hoffmann, Vivian, Christopher B Barrett, and David R Just. 2009. "Do free goods stick to poor households? Experimental evidence on insecticide treated bednets." World Development, 37(3): 607–617.
- Honorine, Solenn. 2010. "Indonesia Hopes to Sell Carbon Credits for Its Forests."
- Jack, B Kelsey, Carolyn Kousky, and Katharine RE Sims. 2008. "Designing payments for ecosystem services: Lessons from previous experience with incentive-based mechanisms." *Proceedings of the National Academy of Sciences*, 105(28): 9465–9470.

JakartaPost. 2013. "Govt plan to lift ban on log exports questioned."

JakartaPost. 2014. "Carbon credit scheme for Indonesia's rainforest launched."

- Jayachandran, Seema. 2009. "Air quality and early-life mortality evidence from Indonesia's wildfires." Journal of Human Resources, 44(4): 916–954.
- Jayachandran, Seema. 2013. "Liquidity Constraints and Deforestation: The Limitations of Payments for Ecosystem Services." *American Economic Review*, 103(3): 309–313.
- **Jensen, John R.** 1996. Introductory digital image processing: a remote sensing perspective. Prentice-Hall Inc.
- Keesing, Felicia, Lisa K Belden, Peter Daszak, Andrew Dobson, C Drew Harvell, Robert D Holt, Peter Hudson, Anna Jolles, Kate E Jones, and Charles E Mitchell. 2010. "Impacts of biodiversity on the emergence and transmission of infectious diseases." Nature, 468(7324): 647–652.
- Koop, Gary, and Lise Tole. 1999. "Is there an environmental Kuznets curve for deforestation?" Journal of Development Economics, 58(1): 231–244.
- Laporta, Gabriel Zorello, Paulo Inácio Knegt Lopez de Prado, Roberto André Kraenkel, Renato Mendes Coutinho, and Maria Anice Mureb Sallum. 2013. "Biodiversity can help prevent malaria outbreaks in tropical forests." *PLoS Neglected Tropical Diseases*, 7(3): e2139.
- Lindsay, Steve W, and Martin Birley. 2004. "Rural development and malaria control in sub-Saharan Africa." *Ecohealth*, 1(2): 129–137.
- Lucas, Adrienne M. 2010. "Malaria eradication and educational attainment: evidence from Paraguay and Sri Lanka." *American Economic Journal: Applied Economics*, 2(2): 46–71.
- Lucas, Adrienne M. 2013. "The impact of malaria eradication on fertility." *Economic Development and Cultural Change*, 61(3): 607–631.
- Margono, Belinda Arunarwati, Peter V Potapov, Svetlana Turubanova, Fred Stolle, and Matthew C Hansen. 2014. "Primary forest cover loss in Indonesia over 2000-2012." Nature Climate Change.
- Miguel, Edward, and Michael Kremer. 2004. "Worms: identifying impacts on education and health in the presence of treatment externalities." *Econometrica*, 72(1): 159–217.
- Myers, Samuel S, Lynne Gaffikin, Christopher D Golden, Richard S Ostfeld, Kent H Redford, Taylor H Ricketts, Will R Turner, and Steven A Osofsky. 2013.
 "Human health impacts of ecosystem alteration." *Proceedings of the National Academy of* Sciences, 110(47): 18753–18760.
- Nature. 2009. "Indonesia to sell carbon credits to conserve forests." Nature, 458(179).
- Neidell, Matthew J. 2004. "Air pollution, health, and socio-economic status: the effect of outdoor air quality on childhood asthma." *Journal of Health Economics*, 23(6): 1209–1236.

- Ngonghala, Calistus N, Mateusz M Pluciński, Megan B Murray, Paul E Farmer, Christopher B Barrett, Donald C Keenan, and Matthew H Bonds. 2014. "Poverty, Disease, and the Ecology of Complex Systems." *PLoS biology*, 12(4): e1001827.
- Pattanayak, Subhrendu K, and Alexander Pfaff. 2009. "Behavior, environment, and health in developing countries: evaluation and valuation." Annual Review of Resource Economics, 1(1): 183–217.
- Pattanayak, Subhrendu, Katherine Dickinson, Catherine Corey, Brian Murray, Erin Sills, and Randall Kramer. 2006. "Deforestation, malaria, and poverty: a call for transdisciplinary research to support the design of cross-sectoral policies." *Sustainability: Science Practice and Policy*, 2(2): 45–56.
- Patz, Jonathan A, and Sarah H Olson. 2006. "Malaria risk and temperature: Influences from global climate change and local land use practices." *Proceedings of the National Academy of Sciences*, 103(15): 5635–5636.
- Patz, Jonathan A, Thaddeus K Graczyk, Nina Geller, and Amy Y Vittor. 2000. "Effects of environmental change on emerging parasitic diseases." *International Journal for Parasitology*, 30(12): 1395–1405.
- **Petney, Trevor N.** 2001. "Environmental, cultural and social changes and their influence on parasite infections." *International Journal for Parasitology*, 31(9): 919–932.
- Pongsiri, Montira J, Joe Roman, Vanessa O Ezenwa, Tony L Goldberg, Hillel S Koren, Stephen C Newbold, Richard S Ostfeld, Subhrendu K Pattanayak, and Daniel J Salkeld. 2009. "Biodiversity loss affects global disease ecology." *Bioscience*, 59(11): 945–954.
- Sachs, Jeffrey, and Pia Malaney. 2002. "The economic and social burden of malaria." *Nature*, 415(6872): 680–685.
- Texier, Gaëtan, Vanessa Machault, Meili Barragti, Jean-Paul Boutin, and Christophe Rogier. 2013. "Environmental determinant of malaria cases among travellers." *Malaria Journal*, 12(1): 87.
- Tucker, Compton J. 1979. "Red and photographic infrared linear combinations for monitoring vegetation." *Remote Sensing of Environment*, 8(2): 127–150.
- Walsh, JF, DH Molyneux, and MH Birley. 1993. "Deforestation: effects on vectorborne disease." *Parasitology*, 106(S1): S55–S75.
- Well, David N. 2007. "Accounting for the effect of health on economic growth." Quarterly Journal of Economics, 122(3): 1265–1306.
- Yasuoka, Junko, and Richard Levins. 2007. "Impact of deforestation and agricultural development on anopheline ecology and malaria epidemiology." *American Journal of Tropical Medicine and Hygiene*, 76(3): 450–460.

Ziebarth, Nicolas, Maike Schmitt, and Martin Karlsson. 2014. "The short-term population health effects of weather and pollution: Implications of climate change." Cornell University.

Tables

| | (1) | (2) | (3) |
|---------------------------------------|-------------|----------------|------------------|
| (Yes = 1, No = 0) | Full Sample | In/Near Forest | Away from Forest |
| | | | |
| Was there an outbreak of malaria? | 0.190 | 0.243 | 0.169 |
| | (0.392) | (0.429) | (0.374) |
| Was there an outbreak of dengue? | 0.0620 | 0.0421 | 0.0702 |
| | (0.241) | (0.201) | (0.255) |
| Was there an outbreak of diarrhea? | 0.176 | 0.204 | 0.164 |
| | (0.380) | (0.403) | (0.370) |
| Was there an outbreak of respiratory? | 0.104 | 0.122 | 0.0971 |
| 1 0 | (0.306) | (0.327) | (0.296) |
| Was there an outbreak of measles? | 0.0806 | 0.0892 | 0.0771 |
| | (0.272) | (0.285) | (0.267) |
| Observations | $143,\!305$ | 41,575 | 101,730 |

Table 1: Fraction of Villages with Disease Outbreaks (Village Level)

Standard deviations in parentheses. Column 2 shows mean and standard deviation for the sub-sample of villages inside or on the border of the forest area. Column 3 shows mean and standard deviation for the sub-sample of villages. These characterizations are made by the village head and verified at the sub-district office. All row-wise differences between columns (2) and (3) are statistically significant.

| | (1) | (2) | (3) |
|--|-------------|----------------|------------------|
| | Full Sample | In/Near Forest | Away from Forest |
| | | | |
| Number of deaths from malaria | 0.873 | 0.746 | 0.945 |
| | (5.275) | (4.498) | (5.670) |
| Number of deaths from dengue | 0.240 | 0.165 | 0.282 |
| | (2.190) | (2.026) | (2.277) |
| Number of deaths from diarrhea | 0.779 | 0.684 | 0.834 |
| | (5.197) | (3.915) | (5.804) |
| Number of deaths from respiratory diseases | 0.665 | 0.492 | 0.765 |
| | (5.958) | (4.767) | (6.541) |
| Number of deaths from measles | 0.341 | 0.260 | 0.387 |
| | (3.036) | (2.639) | (3.240) |
| Observations | 12,454 | 4,531 | 7,923 |

Table 2: Number of Deaths in a village in a given year as a Result of Disease Outbreaks (Village Level)

Standard deviations in parentheses. Column 2 shows mean and standard deviation for the subsample of villages inside or on the border of the forest area. Column 3 shows mean and standard deviation for the sub-sample of villages. All three columns only report averages from the sample of villages that reported an outbreak of that particular disease. All row-wise differences between columns (2) and (3) are statistically significant.

| Table 3: Average Forest Cover | and Deforestat | tion by Forest Zone (Distr | ict Level) |
|---|----------------------------------|--|--|
| | (1) Full Sample | (2) Production/Conversion | (3) Protection/Conservation |
| Deforestation (1000 Ha) - All Districts | 0.724 | 1.269 | 0.199 |
| Ν | (2.021) 9,352 | 3,280 | 3,672 |
| Deforestation (1000 Ha) - Deforested Districts | 0.986 | 1.671 | 0.310 |
| Ν | (3.253) $6,864$ | (4.520) 2,491 2,491 | (1.202) 2,359 |
| Forest Cover (1000 Ha) | 113.6 (206.6) | 160.7 (979.6) | 90.36 (155.5) |
| Ν | 10,521 | 3,690 | 4,131 |
| Sample means by district in 1000s of hectares. Second For more details see Burgess et al. (2012). All | Standard devia row-wise diffe | tion in parentheses follow rences between columns (| ed by sample observations. 2) and (3) are statistically |

significant.

| | (1) | (2) | (3) |
|--|-------------|----------------|------------------|
| | Full Sample | In/Near Forest | Away from Forest |
| | | | |
| 100s of Families Receiving Poor Status Papers | 0.308 | 0.226 | 0.342 |
| | (0.779) | (0.674) | (0.815) |
| 100s of Families Receiving Health Cards | 0.792 | 0.600 | 0.870 |
| | (1.404) | (1.071) | (1.511) |
| Hospital in Village? (Yes $= 1$; No $= 0$) | 0.485 | 0.419 | 0.512 |
| | (0.853) | (0.813) | (0.868) |
| Ease of Access to Nearest Hospital ^a | 2.610 | 3.002 | 2.453 |
| | (0.888) | (0.840) | (0.857) |
| Population Density | 0.00650 | 0.00142 | 0.00855 |
| (1000s of person per hectare) | (0.0334) | (0.00788) | (0.0391) |
| Rice Field Area (1000 Ha) | 0.242 | 0.279 | 0.228 |
| | (1.133) | (1.476) | (0.960) |
| Is the village by a river? (Yes $= 1$, No $= 0$) | 0.711 | 0.801 | 0.675 |
| | (0.453) | (0.399) | (0.468) |
| Elevation Above Sea Level (100 meters) | 1.866 | 2.455 | 1.629 |
| | (4.589) | (5.815) | (3.964) |
| Mean Annual Rainfall (1000mm) ^b | 0.0697 | 0.0728 | 0.0684 |
| | (0.0172) | (0.0184) | (0.0165) |
| Standard Deviation in Rainfall $(1000 \text{mm})^{\text{b}}$ | 0.0315 | 0.0305 | 0.0319 |
| | (0.00956) | (0.00901) | (0.00974) |
| Observations | 120 676 | 29 197 | 04 540 |
| | 152,070 | 30,127 | 94,049 |

Table 4: Key Independent Variables (Village Level)

Standard deviations in parentheses. Column 2 shows mean and standard deviation for the subsample of villages inside or on the border of the forest area. Column 3 shows mean and standard deviation for the sub-sample of villages. These characterizations are made by the village head and verified at the sub-district office.

^a Ease of access defined as follows. 0 = hospital in village, 1 = very easy, 2 = easy, 3 = difficult and 4 = very difficult. These delineations are made by the village head and verified at the district/sub-district office.

^b Rainfall variables calculated over the sample range of 2001-2008.

| Was there an outbreak of Malaria in the Previous Year? | (1) | (2) | (3) | (4) |
|---|--|---|---|--|
| Total Deforestation in the Previous Year (1000 Ha) | 0.00203^{***} (0.000607) | 0.00319^{**} (0.00139) | 0.00287^{**} (0.00122) | 0.00364^{**} (0.00154) |
| One Lead, One Lag District Fixed Effects Island X Year Fixed Effects Island X Year Fixed Effects Distance to Cities, Families on Welfare Agricultural Production, Forest Cover Medical Facilities Access, Terrain, Geography Sanitation, Trash Disposal, Fuel Source and Rainfall Forest Zone, Forest Zone x Province, Forest Zone x Year FE Income Source, x Province FE, x Year FE | X X | XXX | $\times \times \times \times \times$ | $\times \times \times \times \times \times \times \times \times$ |
| Observations R-squared | $ \begin{array}{c} 111,201\\ 0.125 \end{array} $ | $111,201 \\ 0.125$ | $\begin{array}{c} 100,570\\ 0.128\end{array}$ | $100,570 \\ 0.140$ |
| Standard errors clustered at the district level in parent <0.01 , ** p <0.05 , * p <0.1 . Columns 1 and 2 include fixed effects Column 3 includes limited controls and co | heses. Statis e no controls humn 4 inclus | tical significs other than d les the full s | ance is denot listrict and i | ted as *** p sland x year column 1 |

includes only the previous year deforestation variables. Columns 2-4 include 3 time periods of deforestation variables.

| Outbreak |
|-----------------|
| Ĩ. |
| ല |
| aria |
| Ľ, |
| Ц |
| \geq |
| ч |
| Ö |
| Probability |
| on the |
| Deforestation a |
| Η |
| of |
| Impact |
| |
| Table 5 |

| 1 able 0: 1 mpact of Deforestation on the | Frobability | or Uutbrea | LK OT LUTTER | ent Diseases | |
|---|-----------------------------|------------------------------|------------------------------|---------------------------------|----------------------------|
| Was there an outbreak of disease in the previous year? | (1) Malaria | (2) Dengue | (3) Diarrhea | (4) Respiratory | (5) Measles |
| Total Deforestation in the Previous Year (1000 Ha) | 0.00364^{**} (0.00154) | 0.00102 (0.00101) | 0.00125 (0.00146) | 0.00127 (0.00151) | 0.00119 (0.00156) |
| Observations R-squared | $100,570 \\ 0.140$ | 100,570 0.141 | 100,570 0.136 | 100,570 0.136 | $100,570 \\ 0.136$ |
| Standard errors clustered at the district level in paren: $<0.05 + n < 0.1$ All sneeifications include the full range | theses. Stati | stical signifi in columns | cance is der (4) of table | noted as *** p 5 The unit of | <0.01, ** p observation |

| | seases | |
|-------------|---|---|
| ċ | 5 | |
| - | ent I | |
| ٤ | ltter | |
| f | ב | |
| د | 5 | |
| - | eak | |
| - | CDL | |
| (| E C E | |
| c | 5 | |
| | \geq | |
| - | E | |
| · | 01 | |
| | සි | |
| - | 2 | |
| | 9 | |
| | Ы | |
| ſ | 님 | |
| | the Fr | |
| | on the Fr | |
| С - | ation on the Fr | |
| f - - | restation on the Γr | |
| | Jeforestation on the Fr | |
| | Deforestation on the Pr | |
| | of Deforestation on the Pr | |
| | pact of Deforestation on the Pr | |
| | Impact of Deforestation on the Pr | - |
| | to: Impact of Deforestation on the Pr | |
| | de 0: Impact of Deforestation on the Pr | |
| | able 0: Impact of Deforestation on the Fr | |

_ **ODSETVATION** nint of . . Caule 5 Ŧ COLUMNITIES COLUMN COLUCION III ULLE TULL FALLES OF < 0.05, * p < 0.1. All specifications include is village-forestzone-year.

| Table 7: Impact of Deforestation on the | Probability of Malarial | Outbreak: Ro | bustness Checks |
|--|---|---------------------------------------|--|
| Was there an outbreak of Malaria in the Previous Year? | (1) Log Ban Enforcement | (2) Falsification | (3) District Splitting Moratorium |
| Total Deforestation in the Previous Year (1000 Ha) | -0.00139 | 0.00412^{**} | 0.000525 |
| Deforestation in Previous Year X Log Ban Enforcement | (601000) (601000) | (06100.0) | (0.00202) |
| Deforestation in Previous Year X False Log Ban | (16700.0) | 0.000305 | |
| Deforestation in Previous Year X Moratorium Start | | (0.00304) | 0.00632*** |
| Deforestation in Previous Year X Moratorium End | | | (0.00234) - 0.00239 (0.00338) |
| Observations | 74,691 | 74,691 | 74,691 |
| R-squared | 0.145 | 0.145 | 0.145 |
| Standard errors clustered at the district level in parer $* p < 0.1$. All columns include full set of controls in c | theses. Statistical signification of the table 5. In | cance is denot column (1), | ed as *** p <0.01, ** p <0.05, deforestation two vears ago and |
| deforestation two years ago interacted with logging ban and the sum of the two is statistically indistinguishable | enforcement were both st e from zero. That result is | tatistically sign s available in t | nificant but in opposite direction the appendix. |
|) | | | |

| Ē |
|---|
| 5 |
| Robustness (|
| eak: |
| utbr |
| \circ |
| lalarial |
| \geq |
| Эf |
| bability o |
| \sim |
| ro |
| Pro |
| the Pro |
| on the Pro |
| restation on the Pro |
| Deforestation on the Pro |
| of Deforestation on the Pro |
| Impact of Deforestation on the Pro |
| 7: Impact of Deforestation on the Pro |
| ole 7: Impact of Deforestation on the Pro |
| Table 7: Impact of Deforestation on the Pro |

| TADIE O. HIIPAUL OI DEIDIESLAUDII OII LIIE I LUDAUIILY U | N UUUDEAN UL INIAIALIA. T | An entreman incention of | | CALLER ON TOTAL |
|--|-------------------------------|---------------------------------|----------------------------|-------------------------|
| | (1) | | | (4) |
| Was there an outbreak of Malaria in the Previous Year? | Conservation/Protection | Production/Conversion | Outside Forest | Inside/Near Forest |
| Total Deforestation in the Previous Year (1000 Ha) | 0.0136^{***} | 0.00149 | 0.00291 | 0.00647^{***} |
| | (0.00274) | (0.00183) | (0.00212) | (0.00193) |
| Observations | 40,938 | 33,753 | 53,070 | 21,621 |
| R-squared | 0.142 | 0.170 | 0.146 | 0.180 |
| Standard errors clustered at the district level in parenthe $\frac{1}{10000000000000000000000000000000000$ | ses. Statistical significance | is denoted as *** p <0.01, | $^{**} p < 0.05, ^{*} p <$ | 0.1. All specifications |
| include the full range of controls in columns (4) of table | e 5. The unit of observation | n is village-forest zone-yea | r. | |

| iity to Forests | |
|-------------------------------|--|
| one Proxin | |
| sults by Forest Z | |
| Disaggregated Re | |
| of Malaria: | |
| n the Probability of Outbreak | |
| Impact of Deforestation or | |
| Table 8: | |

42

| is denoted | ignificance | Statistical s | entheses. S | Standard errors clustered at the district level in par- |
|------------|-------------|---------------|-------------|---|
| 0.184 | 0.153 | 0.143 | 0.143 | R-squared |
| 100,570 | 100,570 | 111,201 | 111,201 | Observations |
| (0.0258) | (0.0238) | (0.0328) | (0.0203) | |
| 0.00882 | 0.0172 | 0.00443 | 0.0295 | Total Deforestation in the Previous Year (1000 Ha) |
| FE | FΕ | FE | FЕ | Number of people who died from malaria |
| (4) | (3) | (2) | (1) | |
| | Deaths | a Kelated | he Malari | Table 9: Impact of Deforestation on t |

5 ſ -٢ • -Ļ -J. ſ, Ĺ F Ċ Table

as *** p <0.01, ** p <0.05, * p <0.1. Columns (1) - (4) use the same set of controls as Columns (1) - (4) respectively of table 5.

A Appendix

A.1 Descriptives of "Outbreak" of malaria

The following is a kernel density plot of the number of infected persons in a village that declared an outbreak in 2008. The range is 1 to 98.



Figure 1: Kernel Density Distribution of Infections conditional on reported outbreak in 2008

A.2 Choice of Estimator - LPM v. Logit, Probit

The central results are robust to the choice of estimator. Typically, coefficients on logit and probit estimates are 4 and 2.5 times those from the linear probability models. Table A.1 shows that the magnitudes are to the order 8.5 and 4.75 for logit and probit respectively. That means that at best, the linear probability model provides an underestimate of the true effect of deforestation on disease incidence.

Second, I also check to see the percentage of predicted values that lie outside the (0,1) interval. Figure 1 shows the kernel density plot. Only 4% of all predicted observations fell outside the range. My results remain robust to the exclusion of those data points as shown in table A.2.

| | (1) I DM | (2) | (3) |
|--|--|--|----------------------------|
| was there an outbreak of majaria in the previous year. | LPM | LOGIU | Proble |
| Total Deforestation in the Previous Year (1000 Ha) | $\begin{array}{c} 0.00287^{**} \\ (0.00122) \end{array}$ | $\begin{array}{c} 0.0246^{***} \\ (0.00951) \end{array}$ | 0.0136^{**} (0.00530) |
| Observations | $100,\!570$ | 100,262 | 100,262 |

Table A.1: Comparison of Results from LPM, Logit and Probit

Standard errors clustered at the district level in parentheses. Statistical significance is denoted as *** p <0.01, ** p <0.05, * p <0.1. All specifications include the limited range of controls in column (3) of table 5. I don't use the full set of controls in order to reduce computational time associated with the convergence of logit/probit estimators when using maximum likelihood. The unit of observation is village-forestzone-year.

| | (1) | (2) |
|--|--|---|
| Was there an outbreak of malaria in the previous year? | Full Sample | Limited Sample |
| Total Deforestation in the Previous Year (1000 Ha) | $\begin{array}{c} 0.00287^{**} \\ (0.00122) \end{array}$ | $\begin{array}{c} 0.00337^{***} \\ (0.00122) \end{array}$ |
| Observations | $100,\!570$ | $96{,}536$ |
| R-squared | 0.128 | 0.122 |

Table A.2: Linear Probability Model: Full Sample v. Limited Sample

Standard errors clustered at the district level in parentheses. Statistical significance is denoted as *** p <0.01, ** p <0.05, * p <0.1. All specifications include the limited range of controls in column (3) of table 5. Limited sample column (2) results are obtained by reestimating equation in column (1) after dropping observations whose predicted values lie outside the feasible [0,1] range. The unit of observation is village-forestzone-year.

A.3 Linear v. Non-Linear Specification

In this section, I check the robustness of my linear estimation results to non-linear specifications. There is no evidence to support the presence of non-linearities. This could be on account of the relatively short time frame over which we estimate our results.



Figure 2: Kernel Density Distribution of Predicted Outbreaks: Linear Probability Model

| | (1) | (2) | (3) |
|--|----------------|----------------------|-------------|
| Was there an outbreak of malaria in the previous year? | Linear | Linear Approximation | Quadratic |
| | | | |
| Total Deforestation in the Previous Year (1000 Ha) | 0.00364^{**} | 0.00365^{**} | 0.00384 |
| | (0.00154) | (0.00153) | (0.00479) |
| Any Deforestation in Previous Year? | | 0.00294 | |
| | | (0.0124) | |
| Deforestation in Previous Year - Squared | | | -0.0000076 |
| | | | (0.000155) |
| Constant | 0.547*** | 0.548*** | 0.547*** |
| | (0.142) | (0.142) | (0.142) |
| | | | |
| Observations | $100,\!570$ | $100,\!570$ | $100,\!570$ |
| R-squared | 0.140 | 0.140 | 0.140 |

Table A.3: Testing for presence of non-linearities

Standard errors clustered at the district level in parentheses. Statistical significance is denoted as *** p <0.01, ** p <0.05, * p <0.1. All specifications include the limited range of controls in column (4) of table 5. Column (1) is the same as Column (4) of 5. Column (2) includes a dummy variable = 1 if there was any deforestation in the previous year. Column (3) includes a quadratic term. The unit of observation is village-forestzone-year.

A.4 Malaria Related Deaths

In this section, I discuss the robustness of the results in equation (4). In particular, I provide 2 forms of robustness checks: 1) Using deaths per thousands of persons as the dependent variable 2) estimating (4) as a poisson regression and a negative binomial regression.

A.4.1 Cases per thousands of Persons

As shown below the results are qualitatively similar to table 9, in that conditional on a malarial outbreak, there is no effect of deforestation on malaria related deaths.

| | 1 1 | | 1 | |
|--|------------|---------------|---------------|---------------|
| | (1) | (2) | (3) | (4) |
| Number of Malaria Deaths Per Thousand People | ${ m FE}$ | FE | \mathbf{FE} | \mathbf{FE} |
| | | | | |
| Total Deforestation in the Previous Year (1000 Ha) | 8.66e-08 | 6.53e-08 | 1.63e-07 | 1.82e-07 |
| | (1.62e-07) | (9.59e-08) | (1.11e-07) | (1.24e-07) |
| Total Deforestation Two Years Ago (1000 Ha) | -7.52e-08 | | -2.19e-07 | 1.62e-08 |
| | (2.51e-07) | | (1.68e-07) | (1.63e-07) |
| | | | | |
| Observations | $21,\!482$ | $21,\!482$ | 19,088 | 19,088 |
| R-squared | 0.238 | 0.237 | 0.235 | 0.262 |

Table A.4: Using "Number of Deaths per thousand people" as the dependent variable

Standard errors clustered at the district level in parentheses. Statistical significance is denoted as *** p < 0.01, ** p < 0.05, * p < 0.1. Columns (1) - (4) use the same set of controls as Columns (1) - (4) respectively of table 5.

A.4.2 OLS v. Poisson, Negative Binomial

I demonstrate below the robustness of equation (4) to poisson and negative binomial regressions. The poisson and negative binomial results are interpreted as semi-elasticities. The key result doesn't change with the specification - that conditional on a village having an outbreak of malaria, deforestation has no impact on malaria related deaths.

| | (1) | (2) |
|--|---------------------|---------------------|
| Number of people who died from malaria | Poisson | Negative Binomial |
| Total Deforestation in the Previous Year (1000 Ha) | 0.00891 (0.0208) | 0.00417 (0.0104) |
| Observations | 99,297 | 100,570 |

Table A.5: OLS v. Poisson, Negative Binomial

Standard errors clustered at the district level in parentheses. Statistical significance is denoted as *** p <0.01, ** p <0.05, * p <0.1. All columns use limited controls as in column (3) of table 5. The poisson and negative binomial coefficients are interpreted as semi-elasticities.

A.5Back of the envelope calculations

In this section, I describe the assumptions used in making back of the envelope calculations.

Probability of Outbreak to Infected Individuals A.5.1

The baseline result is that 1000 hectares of deforestation increases malarial outbreak by 2% in every village in that district. An outbreak, on average, means that 15 individuals are infected. There are 250 villages, on average, in each district. That means 1000 hectares of deforestation results in 0.02x15x250 = 75 additional infected individuals in each district. Between 2001 and 2008, Indonesia lost 600,000 hectares of forest cover each year. That translates to 45,000 additional infected individuals each year. The same calculations can be repeated at the upper bound estimate of 4.6%, which translates to 110,00 additional infected individuals each year.

Morbidity Cost of Deforestation A.5.2

A single bout of malaria costs 10-20 working days ¹⁸. That means 1000 hectares translates to 750-1500 sick days. I convert this into sick lives by estimating that individuals have a working life of 65 years. I then multiply the resulting number of sick lives with the value of a statistical life (\$3,700,000). I then divide by 1000 to get the per-hectare morbidity cost associated with deforestation-induced malaria. The estimates are 123 - 610 per hectare.

¹⁸See: http://malaria.jhsph.edu/about_malaria/